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Abstract; Compared with hot desulfurization process, white eggshell and granular activated carbon were
used as catalysts to study the influencing factors of light naphtha desulfurization, and the optimum conditions of
light naphtha desulfurization were determined by using a variety of design parameters. The influencing factors of
light naphtha desulfurization mainly include the ratio of hydrogen peroxide and light naphtha, pH value of
solution, stirring speed, temperature, contact time and catalyst dosage. The results show that the conversion of
thermal desulfurization is low, while that of granular activated carbon is high. The desulfurization efficiency is
inversely proportional to the pH of the solution and in direct proportion to other influencing factors. The catalytic
desulfurization method is suitable, economical and environmental friendly to achieve zero residue level. The
results can provide basis for the selection of light naphtha desulfurization methods.
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Tab.1 Characteristic parameters of white eggshell
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6.38 0. 802 1. 075 0.254 0.25~0.50 0.253 25.50
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Tab.2 Characteristic parameters of granular activated carbon
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1.1.3 BERAM#HESS PR T B 5 5 AR T 2B S A AR AL, R TR 22

AL 2R T 00 52 5 A il il A el 28Ok B R B TR TR
PRI T o 2 A Bl L EE (£ 15.6 C AT N
0.692, &t 14 x10 ~° /AR K 51 °C,

1.2 Hik
1.2.1 BmIZiReRE

JBRHRE T 25 32 2 43 Sy Bl PRk ) I B S S o
FE ST ORI P e PR AR A SR AL AR P 28

78 250 mL BB A 100 mL 52 5 43 % i i 4,
FFLA—E e A S8 A S TR, R T R R0 25 15 1K
AR . W T8 55 K W pH E, JF 5 i 4
WEAMEAEN], B L] Ak & mi AL & Y i R, R
A DAR e & A 1 0 A R e K AR 2
IR P R B AR, DT 1 58 5 B Ak B W SR AR it
i rhORH LB PR S I AH AL 5

TE— 7 U BE AN 3K B2 m#OIF R ZR G Y. A

b FUORER N T A 5 AR . A B AN BB i R 5
J& B R O L U8, SR ] ANTEK 9 000 N /S 3 #r4Y
ol X GBI R & . BT S DR
USSR ARG BE , IR I A T 2 R

I A R A B 2 1 2 o S b 5 R B A i v
FIEE (1 ~5) BRI pHAE (1 ~5) JKEFE IR HFE
HEE (100 ~ 500 r /min ) | [ 8 5¢ FE P 4 & (0. 1 ~
2.5 g) JBEAR I EE (25 ~ 85 £0.5 C) Hl4% fih i ] (10 ~
180 min) , 7E 14 X 10 “°HI 1 atm 18 F 400 F , 38 o g 28
IR BB, X AR R R T R B A A A T I RS g
1.2.2 HmEEWNE

R FHAAAG BT A, T3 AS 5] L5 1 18 53 B J0t 580 2% 1Y)
AN .

C. _C,~-C.
%R =_" =

c.” c. (1)

35



36

ATSS0H | 0000 46 A

NATURAL GAS AND OIL

o R BT AL 55 € o IR =22 117 D 3 2 43 v i
WIMRHR BE  C | R ik 78 b Sy 28 43 TR i AR R B2 5 €,
Sk B =2 i D A 4 R R e A B AR

K HI ANTEK 9 000 N /S 23 B AU F X 5 e 2 S AL I
EFIEEE KR ASTM D — 4294 1124 T 500 x 10 I
155 500 X 10 ™ fh B A 2k 2 v JECIM AR 2 P B B o

2 ANmMELHEE I ZENRIME RS
SR IR BH K ik 25 B AR oA it v B9 B8, 0 A FA
TR AN AR T 2 RR
2.1 FRUESERAMMLE
FPE L AT B O AT S A A T ) L
i, BEBR AR B 2 4 i, (HR 2 BB AR T 20 A% L
IRF 3, EARBURR T2 A LU (B TR 5 4 B33 i et 32 el /)
W I T 2R AR

BB
LREE: v

2..50 3..50 4.50
U SR R L

RS 52N b E X AR R B 0
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Fig.2 Effect of pH value on desulfurization efficiency
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Fig.3 Effect of stirring speed on desulfurization efficiency
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Fig.4 Effect of temperature on desulfurization efficiency
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Fig.5 Effect of contact time on desulfurization efficiency
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Fig.6  Effect of catalyst dosage on desulfurization efficiency
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